Author’s response to reviews

Title: Anti-inflammatory activity of edible oyster mushroom is mediated through the inhibition of NF-kB and AP-1 signaling.

Authors:

Andrej Jedinak (Andrej.Jedinak@childrens.harvard.edu)
Shailesh Dudhgaonkar (sdudhgaonkar@gmail.com)
Qing-li Wu (qlwu@aesop.rutgers.edu)
Jim Simon (jimsimon@aesop.rutgers.edu)
Daniel Sliva (dsliva@clarian.org)

Version: 3 Date: 11 April 2011

Author’s response to reviews: see over
April 11, 2011

Dr. Nehme Gabriel
Editor-in-Chief
Nutrition Journal

RE: MS 1053030903468549

Dear Dr. Gabriel,

Please find enclosed our revised manuscript entitled “Anti-inflammatory activity of edible oyster mushroom is mediated through the inhibition of NF-kB and AP-1 signaling.” by Andrej Jedinak, Shailesh Dudhgaonkar, Qing-li Wu, James Simon, and Daniel Sliva.

All requested formatting changes are included in the revised manuscript.

Yours sincerely,

Daniel Sliva, Ph.D.
Cancer Research Laboratory
Methodist Research Institute
Indiana University Health